# RESEARCH

**Open Access** 

Journal of Eating Disorders

# Multi-family therapy for bulimia nervosa: a qualitative pilot study of adolescent and family members' experiences



Anabel Escoffié<sup>1</sup>, Natalie Pretorius<sup>2</sup> and Julian Baudinet<sup>1,2\*</sup>

# Abstract

**Background:** Multi-family therapy (MFT-BN) is a new treatment for adolescent bulimia nervosa with emerging empirical support. It extends the bulimia nervosa focussed family therapy model, by offering treatment in a group setting. Up to nine families work together with a team of clinicians over the course of 20 weeks. No qualitative study to date has investigated the experience of MFT-BN. This study aimed to explore this from the adolescent and parent/ caregiver perspective.

**Methods:** Participants from two consecutive MFT-BN groups facilitated at the Maudsley Hospital in London, UK, were invited to participate in either a focus group or individual qualitative interview about the experience of MFT-BN. Of the 19 eligible participants (from 9 families), 15 (8 parents, 1 older sibling, 6 adolescents) consented and participated. Audio-recordings of interviews and focus groups were transcribed verbatim and analysed using reflexive thematic analysis.

**Results:** Three main themes were identified; (1) seeing and being seen, (2) holistic shift, (3) the unspoken. Participants reported overall shifts in cognitions, emotions, behaviours, and relationships both individually and within the family as a result of attending MFT-BN. Meeting other families with similar struggles and sharing experiences, skills and learning helped reduce isolation and promote change. There was also a sense from participants that some things did not, or could not, be spoken about in the group context and that more direct and challenging conversations might have been helpful at times.

**Discussion:** The current study identifies some of the perceived benefits and challenges of MFT-BN. The three themes demonstrate the holistic nature of change that can occur across the treatment, as well as the power and limits of the group therapy setting and process. Further research is needed to explore the experience of MFT-BN and its outcomes across a more diverse range of participants and treatment settings.

**Keywords:** Adolescent, Bulimia nervosa, Eating disorders, Multi-family therapy (MFT), Family therapy for bulimia nervosa (FT-BN), Family based treatment (FBT)

\*Correspondence: Julian.Baudinet@slam.nhs.uk

<sup>1</sup> Institute of Psychiatry, Psychology and Neuroscience (IoPPN), King's College London, De Crespigny Park, Denmark Hill, London SE5 8AZ, UK Full list of author information is available at the end of the article



© The Author(s) 2022. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/A.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/licenses/by/A.0/. The Creative Commons Public Domain Dedicated in a credit line to the data.

# **Plain English Summary**

Multi-family therapy (MFT-BN) is a new treatment for adolescent bulimia nervosa. It involves a group of families (maximum nine) working together with a team of therapists over the course of 20 weeks. This study aimed to explore the experience of MFT-BN from the adolescents and parent/caregiver perspective, as this has not been explored before. Participants from two consecutive MFT-BN groups were invited to participate in either a focus group or individual interview. 8 parents, 1 older sibling, and 6 adolescents participated. Three themes were identified from what they said. Participants described overall shifts in their thoughts, emotions, behaviours, and relationships both individually and within their families, as a result of attending MFT-BN. Meeting other families with similar struggles and sharing experiences, skills and learning helped reduce isolation and promote change. There was also a sense from participants that some things did not, or could not, be spoken about in the group context and that more direct and challenging conversations might have been helpful at times. More research is needed to explore the experience of MFT-BN and its outcomes across a more diverse range of participants and treatment settings.

# Background

Bulimia nervosa (BN) is a serious mental health disorder that typically develops in adolescence [1-3]. Alongside the main eating disorder symptoms (binge eating and compensatory behaviours, body shape and weight concern/distress), people with BN often experience other co-existing difficulties, such as low mood, self-harm, suicidality, emotion dysregulation and impulsivity [2, 4-9]. Family factors, such as high levels of expressed emotion and parental criticism in response to BN symptoms have also been reported at higher rates compared to parents of young people with anorexia nervosa (AN) or depression [10].

Despite the significant impact of BN on young people and their caregivers, existing research into treatments specifically for young people with BN is limited. BN focussed family therapy (FT-BN) is a recommended treatment for young people in the UK [11], however, outcomes are relatively modest. The NICE (2017) recommendation of FT-BN is largely based on three randomized controlled trials (RCTs) [12–14], in which fewer than 50% of participants were abstinent from bingeing and purging by the end of treatments [12–14]. Clearly, further developments are needed to support both young people with BN and their family members/caregivers.

Multi family therapy (MFT) groups have been developed and used across a number of different physical and mental health disorders for decades [15, 16]. MFT typically involves bringing a group of families together to work with a clinical team over the course of treatment. MFT is usually very activity-based and uses different group constellations (e.g. separated young person, parent and sibling groups, pairs, or mixed groups) to promote sharing and new learning.

In the UK, MFT is now a recommended treatment for adolescents with AN (MFT-AN) [11] and has been manualised for both adolescents [17] and young adults [18]. MFT-AN interventions can be useful as a stand-alone treatment, or as an adjunct treatment alongside other interventions [19]. In the only adolescent outpatient RCT to include MFT-AN, Eisler et al. [20] showed that offering adjunctive MFT-AN alongside family therapy for AN (FT-AN) is associated with better outcomes than FT-AN alone.

Qualitative research into the experience of MFT-AN has identified several benefits afforded from the unique group context [19]. Patients and caregivers describe valuing being able to hear new perspectives, share their own experiences, and learning new skills [21–23]. Nevertheless, MFT can also be described as challenging. Some families have reported concerns about young people mixing in the group and sharing unhelpful eating disorder behaviours, especially if they are at different stages of the illness or recovery [24].

The multi-family group settings may provide additional benefits for adolescents with BN. For example, the group setting might be helpful in providing a supportive and validating context in which to address secrecy and shame (factors often associated with BN), to learn new skills to manage BN and other co-morbid symptoms, and to address family relationships. Young people with BN might also benefit from the social interaction and social support which has been linked to both predisposing and maintenance factors of the illness, and recovery in eating disorders [25, 26].

In an attempt to improve upon outcomes for young people with BN, MFT has recently been adapted specifically for this group (MFT-BN). To date, there has been very little research into MFT interventions for BN; indeed, only one peer review article has been published [19]. In an uncontrolled case series, Stewart et al. [27] found that MFT-BN for adolescents was associated with improvements in bingeing and purging behaviours, comorbid mood, anxiety and emotional regulation difficulties, and parent/caregiver perceived burden. The authors concluded that the intervention could play an important role as a "first step" towards recovery.

While promising, much more research is needed to replicate these findings and to understand the process of treatment. To begin to address this gap, the current study aimed to explore the qualitative experiences of adolescents and their parents/caregivers of MFT-BN.

# Method

# Participants

A convenience sample of adolescents in treatment at the Maudsley Centre for Child and Adolescent Eating Disorders (MCCAED), a specialist, community based, child and adolescent eating disorder service based at the Maudsley Hospital, UK, was used in this study. Adolescents and their family members were eligible for this study if they were (a) between 13 and 18 years old, (b) had an ICD-10 [28] diagnosis of BN or Atypical-BN, (c) were living at home with at least one parent/caregiver and (d) were currently in treatment at MCCAED. ICD-10 diagnoses were made at assessment with MCCAED during clinical interview with an experienced specialist eating disorder clinician. All participants were approached either face-to-face or via telephone contact to take part in the study by a member of the clinical team. Written informed consent was obtained from all participants who took part in this study. Parental consent was obtained in addition to young person assent for those aged below 16 years.

# Description of the MFT-BN intervention and setting

MFT-BN was adapted from an established MFT-AN model [17]. MFT-BN is based on the same principles as the single-family therapy model for BN [29], but extends it by bringing families together to increase support, understanding, and communication. There is also a strong focus on providing practical skills to manage BN symptoms and related co-morbid difficulties. Each MFT-BN group is closed and consists of four to nine families. Participants are offered a total of  $16 \times 90$ -min sessions over the course of 20 weeks. The initial 12 sessions are offered weekly, with the remaining four offered fortnightly. MFT-BN integrates ideas from different therapeutic models, including systemic family therapy, cognitive behavioural therapy and dialectical behaviour therapy. The MFT-BN model has been described in detail elsewhere, including a session-by-session plan [30] and preliminary findings published [27]. Each group varies slightly in content and structure to meet the needs of each specific MFT-BN cohort. The order in which content is delivered is also flexible and can be adapted as needed.

## Study design

Information about the qualitative experiences of MFT-BN was gathered using a combination of focus groups (~60 m) and individual interviews (~45 m), conducted face-to-face in the clinic be author AE alone. All were semi-structured and followed similar topic guides, that focused on (a) experience and expectations of treatment, (b) reflections on what felt more and less helpful in supporting recovery, (c) the impact of MFT-BN on specific domains (see supplementary material for topic guides). Collecting data in different formats was done intentionally with the aim of increasing data richness [31]. All focus groups and interviews were audiorecorded and transcribed verbatim.

## Data analysis

Transcripts of the interviews and focus groups were analysed using reflexive thematic analysis based on Braun and Clarke's [32, 33] and Boyatzis' [34] guidelines. All authors (AE, NP, JB) participated in data analysis and approached it form a critical realist epistemological position, from which meaning and experiences are considered subjective and influenced by social and cultural context. The analysis was more theoretical than inductive as it was driven by FT-BN theory around hypothesised change mechanisms. Themes were identified on a semantic level and the process of analysis was recursive as the researchers went back and forth between the data, codes and candidate themes before arriving at the final findings.

Each data set (i.e. focus groups with parents/caregivers, focus groups with adolescents, interviews with parents/caregivers and interviews with adolescents) was initially analysed independently from the others by author AE. Summaries were then compared to generate initial codes and a codebook for each data set. These were then applied to a sub-sample of the data to check applicability. Codes were then analysed for common overarching themes and sub-themes that could be derived across the different data sets by all three authors (AE, NP, JB). To ensure reliability an independent researcher also reviewed the raw data, initial codes, codebooks and themes at each step of the process. Any discrepancies were resolved through an iterative process, during which the analysing researchers would go back-and-forth between reviewing the data and discussions as a small group about impressions, codes and, eventually, themes. The concept of data saturation was not used during the analysis as it was recognised that different meanings are generated by different researchers and the process is inescapably subjective [35]. No software was used to assist analysis.

# **Researcher characteristics**

First author AE, an MSc student at the time of data collection, was on clinical placement at MCCAED as part of her family therapy training. NP and JB are both clinical psychologists, trained to the doctoral level. Both work within MCCAED and have extensive clinical experience of working with young people with eating disorders and their families. Both have been involved in developing MFT-AN and MFT-BN treatments and clinically facilitate both frequently. Having mixture of genders (female: AE, NP; male: JB), discipline (family therapy, clinical psychology) and level of experience (MSc: AE, Doctoral: NP, JB) was seen as a benefit in the analysis process, as it provided multiple, unique perspectives and meaning on eating disorders, adolescent development and perspectives on treatments.

# Results

# Sample

Participants from two consecutive MFT-BN groups were invited to participate in this study. The total eligible sample included 19 people (7 mothers, 2 fathers, 1 older sibling, 9 adolescents) from 9 families (group 1, n=5 families; group 2, n=4 families). Fifteen (78.9%) of these (6 mothers, 2 fathers, 1 older sibling, 6 adolescents) consented and participated. No specific reasons were

provided by those who did not participate. Six of the parents (4 mothers, 2 fathers) and 3 adolescents provided data via focus groups. Separate groups were facilitated for parents/caregivers versus adolescents. Two mothers, one older sibling and three adolescents completed individual interviews.

Adolescents who participated in this study were aged between 13 and 17 at assessment (mean=15.17, sd=1.17). Four were diagnosed with BN and two with Atypical BN. Five identified as female and one as male. All were cis gendered. Two participants identified as Asian/mixed Asian, one mixed race, and three as white British. Mean duration of eating difficulties prior to assessment at MCCAED was 15.40 months (sd=18.26, range=6–48, median=7). To ensure anonymity of the sample, no further demographic information was collected for this study. No demographic information beyond gender was collected for parent/caregivers.

## **Qualitative responses**

Interview data from the focus groups and semi-structured interviews are synthesized and presented below. A total of three themes and five sub-themes were derived (see Table 1).

1. Seeing and being seen

A common experience for everyone, regardless of family role, was finding it helpful being with others

**Table 1** Themes and sub-themes of adolescent and parent/caregiver experiences

Theme	Subtheme	Illustrative quote
1. Seeing and being seen	1a. Reduced sense of isolation	"It has showed me that other people go through something that I thought was just me. If some people say 'when something happens to me I do this', I go 'oh my God! that happens to me also' and its reassuring" (adolescent)
	1b. Learning from and with each other	"It was good listening to other people's ideas because that helped me think of my own things, of what would help me" (adolescent)
2. Holistic shift	2a. Family connection	"My mum and dad [now] know how to help me, as opposed to just getting angry" (adolescent) " I am definitely getting along with them [parents] better" (adolescent)
	2b. New insights and coping	"Like making a set plan of the eating, I thought that was helpful 'cause then like you are not hungry for you to binge or whatever and things like writing down your wor- ries that was helpful" (adolescent) "I think it's benefited her as I said she has some insight into the reasons why she is doing it, so it's definitely had a positive impact on her" (parent/caregiver)
	2c. Practical help	"It's a lot of talking obviously, and then there's things like distractions and they do give you a lot of sheets [coping skills leaflets] so you can take stuff home with you so that it doesn't stay at the session cause it's easy to go home and kind of push it to the side but because they give you sheets and then you go through your bag and then you find it and you kind of remember and that's good" (adolescent)
3. The unspoken	[no subtheme]	"[there] weren't exactly rules but you do start to think 'well is what I'm about to say, ok' and stuff like that, so you do kind of have to think twice before you say stuff which really isn't the point I don't think" (adolescent)

and working towards a common purpose. Participants described feeling less isolated by being around others struggling with similar difficulties and also benefited from being able to learn and share their own experiences and skills.

a. Reduced sense of isolation

Adolescents and parents/caregivers alike described the normalising and validating effects of being in a group context with others who shared the same problem. There was a sense of togetherness within the group, as opposed to feeling lonely in the situation of living with BN.

Parent: "It's helped for us knowing that somebody else is in the same situation. That makes you feel a lot better knowing that. ... I know it's not nice to say, but that you're not isolated, you're not the only one. That makes you feel ten times better ... would recommend that to anyone"

Adolescent: "I think she [parent/caregiver] probably found it extremely helpful that she could speak to other girls my age [with BN] and other parents, you know? Just speaking to other people, so that she doesn't feel alone in it, as I felt alone in it..." (adolescent)

b. Learning from and with each other

Adolescents and parents/caregivers said that being with others was an opportunity to learn by both talking and listening to other peoples' points of views. Participants expressed how being in a group context encouraged perspective taking, which enabled everyone to be more receptive and empathic. This process was described as helpful in both understanding others as well as the self.

Adolescent: "It's good to see other people's opinions. It makes your opinion more rounded and you understand and empathise with more people and then you understand more [for yourself]" Parent: "It gave me a greater understanding of what he [adolescent] was going through because I just didn't understand, I thought what sane person would do that to themselves"

# 2. Holistic shift

Adolescents and parents both expressed that MFT-BN helped everyone to function and cope better in a myriad of ways. Through different mechanisms, such as learning skills, understanding patterns differently, and having space to think more broadly about general mood and well-being, participants described small and large shifts in cognitions, emotions, behaviours and relationships. Some were small, others large. Some were very obvious and intentional/active changes; others were more about acceptance and slowing down.

a. Family connection

Most participants described changes in the way all family members connected as a result of MFT-BN. Participants said that their relationships became closer and more trusting. Others described improvements in communication, as well as in people's awareness of others' emotions.

Adolescent: "I think we [mum and I] get on better now, 'cause like, I think she kind of accepts it now ... accept[s] it in the way that she wouldn't not do anything, but like accepted that this is going on and that having a go at me isn't gonna work, and she is less in my face about it"

Adolescent: "when we started coming here I think it brought us back closer again 'cause, like, I've got more trust in her"

Parent: "I think our [parent-child] relationship has improved, definitely, we don't argue as much which is good"

# b. New insights and coping

While adolescents described multiple shifts across treatment, typically they were more focused on improvements in general wellbeing and coping rather than specific behaviour changes. It was rare for an adolescent to specifically mention BN symptoms. Whereas parents/ caregivers often explicitly noted improvements for the adolescents in terms of a reduction in specific bulimic symptoms, such as bingeing, purging and restriction, and a generally healthier approach to food and eating.

Parent: "She [my child] is quite confident now that she can control it [bingeing] and she tells me that she's been great for quite some time and she is definitely eating much better, and is not a problem for her"

Adolescent: "It's been really helpful, ... its helped, like, [me] cope with things rather than stop eating. 'Case you can't really ever do that, I don't think."

These improvements were often linked to new insights and understanding around common binge/purge triggers, as well as cognitive, emotional, relational and behavioural patterns associated with symptoms of BN and low mood. Parent: "I think for them [the adolescents], they learned like their triggers and what makes them feel in that way and then what makes them go in the vicious cycle"

Adolescent: "... normally you don't think about what's going on and, like, when you have, like, a binge or whatever, you don't think about it after so it's good that we did that to make us see where we want to stop and how we can stop it if that makes sense"

Similarly, several parents/caregivers also specifically spoke about how the adolescents seemed to be managing strong emotions and mood swings more effectively by the end of MFT-BN.

Parent: "I mean, before it was all tears and everything, but now she is much happier, she knows how to look after her body and she wants to set up some type of web that she can help somebody else"

#### c. Practical help

The focus on skills in the programme was specifically described by many as beneficial. It was often directly linked to improved symptom management, especially related to binge/purge behaviours.

Adolescent: "If I'm feeling anxious a good one for me to do is try and put a lot of sensation in my feet [to ground and distract myself], so like walk on grass or like foot massage ... like walk on a funny texture or something like that"

Parent: "She [adolescent] uses all the techniques that were given to them, you know, to break the cycle [of bingeing and purging] ... There are days in which she just you know gives into temptation or whatever, but I think it's benefited her, she is managing to deal with it better than she did before"

Parent: "On the other hand ... in the group as parents, we have been given some skills to cope better in addition too"

# 3. The unspoken

The final theme identified was related to what remained unspoken in MFT-BN. It related, in part, to different people's perceptions of the group structure and process, as well as the role that shame and guilt had on what people felt comfortable to discuss.

From the parent/caregiver's perspective, many reflected that BN symptoms were a difficult subject for adolescents to talk about, especially with their own parents and/or in a group setting. They noted there was something challenging about having to actively addressing emotions such as shame, embarrassment or guilt.

Parent: "... I understand it is a very secretive disorder and, you know, and very embarrassing [to talk about in a group] as well. Especially with all the stigma against it"

From the adolescent perspective, however, several felt the group structure and process inhibited their ability to express themselves at times. Some remarked on avoiding certain topics, or withholding details, particularly around risk issues. This was out of concern that parents or clinicians in the group may 'overreact'. This balance between openness, confidentiality, and risk felt tricky to navigate for some.

Adolescent: "I think it's quite similar to school, you can tell your teachers some things but there are some things that you know you can't, like things that would put you in danger or whatever but then I'm thinking that's the reason we're here is because we have a disorder and that's putting us in danger, so I didn't really understand that and that made me feel like I had to hold back quite a few things, that I couldn't say everything that I wanted to say because I might be reported or whatever"

Others noted that this could limit how direct or challenging a conversation could be. Some felt more directness was needed sometimes to move the conversations to a more meaningful place. The importance of needing to trust each other's intentions and not take things too personally was emphasised.

Adolescent: "It was quite cutesy and I think if they were more harsh and I think if we were all more harsh in what we said, then I think a lot more would come out of it and we would think about what other people said a lot more. I think because we were all so careful about what we said and we were all pussyfooting around other people's feelings. I think if we all made a pact at the beginning like what we say, we say not to be horrible or rude or mean it's just what we are thinking so don't take it offensively"

To address this, several adolescents expressed a desire for more unstructured time, together as an adolescent group, away from the parents. There was a sense of wanting to share more of their week-to-week experiences in a more informal way.

Adolescent: "I think the structure was good, but I think that there should be a few sessions where nothing has been planned and we literally can sit there

and discuss how our weeks have been. I felt sometimes that I just wanted to say if I have had a really crap week. I wanted to tell people how I was feeling and how crap I felt"

## Discussion

The current study aimed to explore the adolescent and the parent/caregiver experience of a 20-week MFT-BN programme. While there is preliminary data suggesting MFT-BN may be an effective intervention for adolescents [27], there is no data, to our knowledge, exploring participant experiences of the treatment.

Overall, three themes were identified from the data; (1) Seeing and being seen (the group as a space for support and learning), (2) holistic shift (a space that facilitated a broad array of changes) and (3) the unspoken (reflections on the scope and limitations of the group process). It was strongly noted by adolescents and parents/caregivers alike that MFT-BN is a space of learning and support. Joining with other families in a similar situation helped participants to feel less isolated, less burdened by the illness and was described as helpful in learning new ways of managing the illness.

Relatedly, multiple levels of change were described for participants. This ranged from shifts in family communication patterns and trust and improvement in symptoms of BN, to insight into their function and broader emotional well-being. These changes were sometimes described differently from the adolescent and the parent/caregiver perspective, but all revolved around a main theme of shifting behaviour, understanding, and relationships. The themes identified in this study add to what is already known about some of the socio-cognitive factors in eating disorders and what might be helpful to target in treatment. For example, the first theme (seeing and being seen) is consistent with previous findings suggesting people with eating disorders value the social interactions of being with others in a group setting [36]. Similarly, the findings of increased family connection and trust reported by participants in theme two (holistic shift) links with suggestions of the importance of treatment needing to address social difficulties which may predispose and maintain eating disorders [25].

Nevertheless, the group identified limits to what was spoken about in the group. Parents/caregivers noted how hard it can be for adolescents with BN to speak about the illness and the associated shame and guilt. Adolescents noted that there were times when they felt the conversation was limited by the group expectations around appropriate information sharing. While the structure was found to be helpful, some expressed a desire to also have some less structured time and to be challenged more. This last theme really reflected the dialectic between needing the containing nature of a structured group programme to contain anxiety, and the need to be flexible in how treatment is delivered.

The MFT-BN programme is intentionally highly structured as a means of containing group anxiety and helping to promote discussion and cohesion. Given uncertainty and anxiety about the group process is usually highest at the start of treatment, perhaps this structure is only needed early on in process and needs to loosen once participants are more engaged in the process.

The current findings are broadly consistent with previous research on MFT for adolescent eating disorders [19]. This literature highlights the group as place of support and learning associated with symptom change [21– 24]. As a point of difference, participants in this study did not mention any unhelpful aspects of the inevitable comparisons that occur between participants in the group. This has been reported by some participants, particularly parents, who receive MFT-AN in other studies [24]. Similarly, participants in MFT-AN studies have not identified wanting to be challenged more in group, as was the case in this study.

This may, in part, reflect differences in the way individuals with BN and their family members present to treatment compared to other eating disorders, such as AN. Additionally, treatment factors may also be contributing to these differences. One important point of difference between MFT-BN and other types of MFT for eating disorders is its intensity. In several MFT-AN models, the group meets for 4 or 5 consecutive full days of intervention [17, 37, 38], which can be followed by additional stand-alone full days of intervention spread over several months [19]. Perhaps participants in the current study felt unchallenged by the group process, at times, simply due to this less intensive format.

The MFT-BN model is intentionally less intensive than those for AN, due the high levels of expressed emotion and criticism often reported within families, as well as the secretive nature of the illness and related emotions of shame and guilt. A briefer, more contained structure with more time devoted to separated adolescent and parent groups is designed to specifically ameliorate some of these factors by ensuring familial patterns are not reenacted in the group. Feedback that the group may not be challenging enough at times is worth considering in future developments of the model.

### Strengths and limitations

This study has several strengths. It is the first study to explore the qualitative outcomes of MFT-BN for adolescents and adds to the limited overall existing research on interventions for adolescent BN. The sample was also relatively diverse, allowing for a mixture of perspectives. Utilising a combination of focus groups and individual interviews is another strength as it increases data

richness. Regarding limitations, the sample size is small and data collected from only two groups facilitated in one service. The current findings need to be replicated in larger and different groups of young people across a range of treatment settings.

# Conclusions

The current findings support the importance of continuing to develop family, and multi-family, interventions for adolescents with BN. Facilitating group spaces, in which families can learn from each other, appears to be a useful way of increasing support for all, as well as promoting a range of important symptom and relational changes. More data is needed to explore MFT-BN outcomes and experiences across a range of different settings. However, initial data is promising and provides new avenues of exploration in a field that continues to be relatively understudied.

#### Abbreviations

AN: Anorexia nervosa; BN: Bulimia nervosa; ICD-10: International Classification of Diseases, 10th edition; FT-AN: Anorexia nervosa focussed family therapy; FT-BN: Bulimia nervosa focussed family therapy; MCCAED: Maudsley Centre for Child and Adolescent Eating Disorders; MFT: Multi-family therapy; MFT-AN: Multi-family therapy for anorexia nervosa; MFT-BN: Multi-family therapy for bulimia nervosa; NICE: National Institute for Health and Care Excellence.

#### Acknowledgements

We wanted to thank all the young people and family members that gave their time to this project. We really appreciate it.

#### Author contributions

AE completed this project as part of a MSc dissertation. NP supervised the work. JB contributed to data analysis and manuscript preparation. All authors contributed to, read, and approved the final manuscript.

#### Funding

There was no external funding for this study.

#### Availability of data and materials

Data are available from the corresponding author on reasonable request.

#### Declarations

#### Ethics approval and consent to participate

This study was approved by the London-Dulwich NHS Research Ethics (REC Reference: 12/LO/0731. Written informed consent was provided by all participants in this study. For those under 16 years of age parental consent was also provided in addition to adolescent assent.

#### **Consent for publication**

Consent for publication was provided as part of written informed consent.

#### **Competing interests**

JB receives royalties from Routledge for a published treatment manual for MFT-AN.

#### Author details

<sup>1</sup>Institute of Psychiatry, Psychology and Neuroscience (IoPPN), King's College London, De Crespigny Park, Denmark Hill, London SE5 8AZ, UK. <sup>2</sup>Maudsley Centre for Child and Adolescent Eating Disorders (MCCAED), Maudsley Hospital, De Crespigny Park, Denmark Hill, London SE5 8AZ, UK.

#### Received: 1 April 2022 Accepted: 12 June 2022 Published online: 03 July 2022

#### References

- Hail L, Le Grange D. Bulimia nervosa in adolescents: prevalence and treatment challenges. Adolesc Health Med Ther. 2018;9:11–6.
- 2. Swanson SA. Prevalence and correlates of eating disorders in adolescents: results from the national comorbidity survey replication adolescent supplement. Arch Gen Psychiatry. 2011;68(7):714.
- van Eeden AE, van Hoeken D, Hoek HW. Incidence, prevalence and mortality of anorexia nervosa and bulimia nervosa. Curr Opin Psychiatry. 2021;34(6):515–24.
- Hovrud L, Simons R, Simons J, Korkow J. Non-suicidal self-injury and bulimia: the role of emotion dysregulation and body dissatisfaction. Eat Weight Disord. 2020;25(4):1089–97.
- Dansky BS, Brewerton TD, Kilpatrick DG. Comorbidity of bulimia nervosa and alcohol use disorders: results from the National Women's Study. Int J Eat Disord. 2000;27(2):180–90.
- Martinussen M, Friborg O, Schmierer P, Kaiser S, Øvergård KT, Neunhoeffer AL, et al. The comorbidity of personality disorders in eating disorders: a meta-analysis. Eat Weight Disord. 2017;22(2):201–9.
- Ulfvebrand S, Birgegård A, Norring C, Högdahl L, von Hausswolff-Juhlin Y. Psychiatric comorbidity in women and men with eating disorders results from a large clinical database. Psychiatry Res. 2015;230(2):294–9.
- Himmerich H, Hotopf M, Shetty H, Schmidt U, Treasure J, Hayes RD, et al. Psychiatric comorbidity as a risk factor for the mortality of people with bulimia nervosa. Soc Psychiatry Psychiatr Epidemiol. 2019;54(7):813–21.
- 9. Miniati M, Benvenuti A, Bologna E, Maglio A, Cotugno B, Massimetti G, et al. Mood spectrum comorbidity in patients with anorexia and bulimia nervosa. Eat Weight Disord. 2018;23(3):305–11.
- Rienecke RD, Sim L, Lock J, Le Grange D. Patterns of expressed emotion in adolescent eating disorders. J Child Psychol Psychiatr. 2016;57(12):1407–13.
- National Institute for Health and Care Excellence (NICE). Eating disorders (NICE Guideline ng69). 2017. Retrieved from: https://www.nice.org.uk/ guidance/ng69.
- Le Grange D, Crosby RD, Rathouz PJ, Leventhal BL. A randomized controlled comparison of family-based treatment and supportive psychotherapy for adolescent bulimia nervosa. Arch Gen Psychiatry. 2007;64(9):1049–56.
- Le Grange D, Lock J, Agras WS, Bryson SW, Jo B. Randomized clinical trial of family-based treatment and cognitive-behavioral therapy for adolescent bulimia nervosa. J Am Acad Child Adolesc Psychiatry. 2015;54(11):886-894.e2.
- Schmidt U, Lee S, Beecham J, Perkins S, Treasure J, Yi I, et al. A randomized controlled trial of family therapy and cognitive behavior therapy guided self-care for adolescents with bulimia nervosa and related disorders. Am J Psychiatry. 2007;164(4):591–8.
- Cook-Darzens S, Gelin Z, Hendrick S. Evidence base for multiple family therapy (MFT) in non-psychiatric conditions and problems: a review (part 2): evidence base for non-psychiatric MFT. J Fam Ther. 2018;40(3):326–43.
- Gelin Z, Cook-Darzens S, Hendrick S. The evidence base for multiple family therapy in psychiatric disorders: a review (part 1). Anderson, editor. J Fam Ther. 2018;40(3):302–25.
- Simic M, Baudinet J, Blessitt E, Wallis A, Eisler I. Multi-family therapy for anorexia nervosa: a treatment manual. 1st ed. London: Routledge; 2021.
- Tantillo M, McGraw JS, Le Grange D. Multifamily therapy group for young adults with anorexia nervosa: reconnecting for recovery. 1st ed. London: Routledge; 2020. p. 324.
- Baudinet J, Eisler I, Dawson L, Simic M, Schmidt U. Multi-family therapy for eating disorders: a systematic scoping review of the quantitative and qualitative findings. Int J Eat Disord. 2021;54(12):2095–120.

- Eisler I, Simic M, Hodsoll J, Asen E, Berelowitz M, Connan F, et al. A pragmatic randomised multi-centre trial of multifamily and single family therapy for adolescent anorexia nervosa. BMC Psychiatry. 2016;16(1):422.
- Engman-Bredvik S, Suarez NC, Levi R, Nilsson K. Multi-family therapy in anorexia nervosa-a qualitative study of parental experiences. Eat Disord J Treat Prev. 2016;24(2):186–97. https://doi.org/10.1080/10640266.2015. 1034053
- 22. Voriadaki T, Simic M, Espie J, Eisler I. Intensive multi-family therapy for adolescent anorexia nervosa: adolescents' and parents' day-to-day experiences. J Fam Ther. 2015;37(1):5–23.
- Wiseman H, Ensoll S, Russouw L, Butler C. Multi-family therapy for young people with anorexia nervosa: clinicians' and carers' perspectives on systemic changes. J Syst Ther. 2019;38(3):67–83.
- Baumas V, Zebdi R, Julien-Sweerts S, Carrot B, Godart N, Minier L, et al. Patients and parents' experience of multi-family therapy for anorexia nervosa: a pilot study. Front Psychol. 2021;12(101550902): 584565.
- Cardi V, Tchanturia K, Treasure J. Premorbid and illness-related social difficulties in eating disorders: an overview of the literature and treatment developments. Curr Neuropharmacol. 2018;16(8):1122–30.
- de Vos JA, LaMarre A, Radstaak M, Bijkerk CA, Bohlmeijer ET, Westerhof GJ. Identifying fundamental criteria for eating disorder recovery: a systematic review and qualitative meta-analysis. J Eat Disord. 2017;5(1):34.
- Stewart CS, Baudinet J, Hall R, Fiskå M, Pretorius N, Voulgari S, et al. Multifamily therapy for bulimia nervosa in adolescence: a pilot study in a community eating disorder service. Eat Disord J Treat Prev. 2019;29(4):351–67.
- World Health Organization, editor. International statistical classification of diseases and related health problems (ICD-10). 2nd ed. Geneva: World Health Organization; 2004.
- Eisler I, Simic M, Blessitt E, Dodge L, MCCAED Team. Maudsley service manual for child and adolescent eating disorders. 2016. https://mccaed. slam.nhs.uk/wp-content/uploads/2019/11/Maudsley-Service-Manualfor-Child-and-Adolescent-Eating-Disorders-July-2016.pdf.
- Stewart C, Voulgari S, Eisler I, Hunt K, Simic M. Multi-family therapy for bulimia nervosa in adolescence. Eat Disord. 2015;23(4):345–55.
- Lambert SD, Loiselle CG. Combining individual interviews and focus groups to enhance data richness. J Adv Nurs. 2008;62(2):228–37.
- Braun V, Clarke V. Using thematic analysis in psychology. Qual Res Psychol. 2006;3(2):77–101.
- Braun V, Clarke V. One size fits all? What counts as quality practice in (reflexive) thematic analysis? Qual Res Psychol. 2020;3:1–25.
- Boyatzis RE. Transforming qualitative information: thematic analysis and code development, vol. xvi. Thousand Oaks, CA: Sage Publications; 1998. (Transforming qualitative information: Thematic analysis and code development.).
- Braun V, Clarke V. To saturate or not to saturate? Questioning data saturation as a useful concept for thematic analysis and sample-size rationales. Qual Res Sport Exerc Health. 2021;13(2):201–16.
- Hagan WS, Mericle S, Hunt BJ, Harper JA, Palka JM, Pelfrey S, et al. Qualitative patient experiences from the self-blame and perspective-taking intervention for eating disorders. J Eat Disord. 2021;9(1):127.
- Knatz S, Murray SB, Matheson B, Boutelle KN, Rockwell R, Eisler I, Kaye WH. A brief, intensive application of multi-family-based treatment for eating disorders. Eat Disord J Treat Prev. 2015;23(4):315–24. https://doi.org/10. 1080/10640266.2015.1042318
- Knatz Peck S, Towne T, Wierenga CE, Hill L, Eisler I, Brown T, et al. Temperament-based treatment for young adults with eating disorders: acceptability and initial efficacy of an intensive, multi-family, parentinvolved treatment. J Eat Disord. 2021;9(1):110.

#### **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

#### Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

#### At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

